Is Audit Committee Equity Compensation Related to Audit Fees?*

XINMING LIU, School of Management, Center for Accounting Studies of Xiamen University, Xiamen University

GERALD J. LOBO, C.T. Bauer College of Business, University of Houston

HUNG-CHAO YU, College of Commerce, National Chengchi University[†]

ABSTRACT

Section 301 of the Sarbanes-Oxley Act (SOX) implicitly assumes that audit committees can independently determine audit fees. Critics of section 301 have questioned this assumption in particular, and the efficacy of section 301 more generally. In response, the SEC issued a concept release in 2015 calling for public disclosure of the process that audit committees follow for determining auditor compensation. Motivated by these calls and the widespread use of stocks and options to compensate firms' independent directors, we examine the relation between equity compensation granted to audit committee members and audit fees. Using a sample of 3,685 firm-year observations during 2007-2015, we find a negative relation between audit committee equity compensation and audit fees, consistent with larger equity pay inducing audit committee members to compromise independence by paying lower audit fees. These findings are robust to controlling for endogeneity, firm size, alternative measures of equity compensation, alternative samples, and an alternative treatment of extreme values. We further show that larger equity compensation is associated with lower earnings quality. We also find that the negative effect of equity compensation on audit fees is stronger when city-level audit market competition is high. However, this negative relation disappears when (i) firms face high litigation risk, (ii) auditors have stronger bargaining power, (iii) the audit committee includes a high proportion of accounting experts, and (iv) auditors are industry experts. Our results are relevant for regulators and investors.

Keywords: audit committee, audit fees, auditor independence, equity compensation

Y a-t-il une relation entre la rémunération en actions des membres d'un comité d'audit et les frais d'audit?

RÉSUMÉ

L'article 301 de la loi Sarbanes-Oxley (SOX) suppose implicitement que les comités d'audit peuvent déterminer les frais d'audit de façon indépendante. Les critiques relatives à l'article 301 ont remis en question cette supposition et, de façon plus générale, l'efficacité de l'article 301. En

^{*} Accepted by Thomas Omer. We thank Thomas Omer, two anonymous reviewers, Tai-Yuan Chen, Ting-Kai Chou, Chris Hogan, Yong-Tae Kim, Elaine Mauldin, Shiva Rajgopal, Michael Willenborg, Hong Xie, and participants at the 2015 AAA Annual Meeting for their valuable comments and suggestions on an earlier version of this paper. Xinming Liu acknowledges the financial support from the National Natural Science Foundation of China (NSFC-71790602 and NSFC-71672159) and Fundamental Research Funds for the Central Universities (Grant number 20720191033). Hung-Chao Yu greatly appreciates the financial support from the Ministry of Science and Technology, ROC (MOST 03-2410-H-004-031). An earlier version of this paper was filed with MOST as a requirement of the funding. Data availability statement: The data used in this paper are available from Audit Analytics, BoardEx, Compustat, and ExecuComp.

[†]Corresponding author.

réponse à ces critiques, la SEC a fait paraître en 2015 un document de consultation demandant que soit rendu public le processus mis en œuvre par les comités d'audit pour déterminer la rémunération des auditeurs. Dans la foulée de ces demandes et de l'utilisation répandue des actions et options pour rémunérer les administrateurs indépendants des sociétés, nous examinons la relation entre la rémunération en actions accordée aux membres des comités d'audit et les frais d'audit. À partir d'un échantillon de 3 685 observations année-entreprises portant sur la période 2007 à 2015, nous constatons une relation négative entre la rémunération en actions des comités d'audit et les frais d'audit, ce qui cadre avec l'hypothèse voulant qu'une rémunération en actions plus élevée incite les membres des comités à compromettre leur indépendance en payant des tarifs d'audit plus faibles. Ces résultats sont robustes pour la prise en compte de l'endogénéité, de la taille des sociétés, d'autres mesures de rémunération par actions, d'autres échantillons et d'une autre méthode de traitement des valeurs extrêmes. Nous établissons également qu'une rémunération en actions plus importante est associée à des résultats de plus faible qualité. Nous constatons enfin que l'effet négatif de la rémunération en actions sur les frais d'audit est plus marqué lorsque la concurrence sur le marché de l'audit à l'échelle de la ville est forte. Toutefois, cette relation négative disparaît lorsque i) les sociétés font face à des risques de litige élevés, ii) les auditeurs disposent d'un plus grand pouvoir de négociation, iii) le comité d'audit comprend une forte proportion d'experts comptables et iv) les auditeurs sont des experts du secteur d'activités. Nos résultats sont pertinents pour les organismes de réglementation et les investisseurs.

Mots clés : comité d'audit, frais d'audit, indépendance des auditeurs, rémunération par actions

1. Introduction

We investigate the relation between audit fees and equity compensation (i.e., stocks and options) granted to audit committee (AC) members. Over the past decades, audit pricing has raised serious concerns because management could determine audit fees in the auditors' favor when it makes the auditor compensation decision (Choi et al. 2010; Kinney and Libby 2002) and, as a result, the auditors could view their role as serving management rather than users of financial statements (SEC 2003) and thus potentially compromise their independence. To address these threats to auditor independence, section 301 of the Sarbanes-Oxley Act (SOX) requires that ACs be directly responsible for the compensation of auditors. By empowering ACs to determine auditor compensation, regulators expect to eliminate management's influence over the auditor and to better align the auditor's incentives with those of the board and shareholders.

Although regulators assume that AC members will independently determine audit fees under section 301 (Doty 2014), there is no post-SOX empirical evidence on whether ACs actually exercise this responsibility in an independent manner. This study examines whether AC independence is affected by equity compensation which, in turn, influences AC members' audit fee decisions. Our research question is important for several reasons. First, if the empirical evidence shows that equity compensation can align the interests of AC members with those of shareholders, then regulators and investors will have confidence in firms' financial reporting quality because AC members will not compromise independence in fulfilling their section 301 responsibilities. If, however, the empirical evidence shows the opposite result, then the requirements under section 301 may instill false confidence in investors that AC members who receive large equity compensation are able to mitigate managerial influence during the auditor hiring and audit fee negotiation processes. Second, governance activists have recently called for greater transparency from ACs regarding how they appoint, compensate, and monitor external auditors (e.g., Audit Committee Collaboration 2013). This interest in transparency is in part due to the concern that, even though the obligations of ACs have expanded over the years, required AC disclosures in this area predate SOX (Ernst & Young 2017). Third, regulators have also noted the possibility that ACs may not be effectively complying with section 301. For example, the SEC issued a concept release in 2015 that proposed new disclosure requirements relating to AC assessments of the auditors. The SEC indicated that such disclosure "may include a description of the nature of the audit committee's involvement in evaluating and approving the auditor's compensation" (SEC 2015, 39) because it helps investors understand and evaluate AC performance (White 2014; Whitehouse 2014).¹ Overall, the activists' reports and SEC (2015) suggest that the efficacy of section 301 has become a regulatory concern and regulators believe that this concern cannot be mitigated if investors do not know how auditor compensation is determined. Given the scarcity of empirical evidence on whether ACs make audit fee decisions independently and the factors that affect their independence, our study attempts to fill this void by addressing these important issues and responding to the calls for more research on the interactions between ACs and audit quality (CAQ 2015; DeFond and Zhang 2014).

We focus on equity compensation because stocks and options, which are widely used to compensate AC members (Engel et al. 2010),² may tie AC members' wealth to a firm's shortterm and long-term financial performance (Millstein 2002). Because the objectivity of AC members in determining appropriate audit fee levels may decrease when their compensation creates conflicts that induce them to compromise independence (Magilke et al. 2009), equity compensation raises a potentially serious concern not addressed by SOX that could harm AC independence. The auditing literature shows that option compensation is associated with a higher likelihood of restatements (e.g., Archambeault and Hermanson 2008), earnings management (e.g., Bedard et al. 2004), beating analyst forecasts (e.g., Campbell et al. 2015), and waiving misstatements to avoid missing earnings forecasts (e.g., Keune and Johnstone 2015). In contrast, the corporate governance literature indicates that it may be beneficial to have outside directors own equity because equity ownership aligns their interests with those of shareholders (e.g., Fama and Jensen 1983; Monks and Minow 2011) and provides incentives for monitoring (e.g., Beasley 1996). In addition, AC members have strong incentives to maintain their reputation due to potential litigation exposure (Naiker and Sharma 2009; Naiker et al. 2013) and discipline by the labor market (Srinivasan 2005). These findings suggest that equity compensation may not affect AC independence because of the potential litigation risk and reputation loss. Overall, the preceding discussion indicates that despite all AC members being required under SOX section 301 to be fully independent, the mixed results reported in the corporate governance and auditing literatures suggest that the net effect of equity compensation on AC independence remains an empirical question. We provide evidence on this relation.

Following Sengupta and Zhang (2015), we use the ratio of the sum of stocks and options to total compensation to examine the effect of AC equity compensation on the quality of firms' financial statements, which is influenced by how effectively ACs determine the audit fee level. Using a sample of 3,685 firm-year observations from 2007 to 2015, we find a negative association between audit fees and AC equity compensation ratio. Our results indicate that a one-unit increase in AC equity compensation ratio is associated with a reduction of \$297,710 in audit fees. This negative association is robust to controlling for endogeneity using a change model, including firm fixed effects, and controlling for omitted time-variant variables. We also find a negative relation between four proxies for earnings quality (i.e., performance-adjusted discretionary accruals, restatement likelihood, and two aggregate measures of real earnings management) and AC equity compensation ratio. Overall, our main results show that a larger equity compensation motivates ACs to choose a lower audit fee level, which, in turn, leads to poorer earnings quality.

This concept release was formally included in the SEC's rulemaking agenda in July 2017, but did not result in major changes to date. Instead, the SEC takes a monitoring approach and encourages public firms to voluntarily disclose their audit committee activities (Austin et al. 2018).

^{2.} We find that the ratio of equity compensation to total compensation ranges between 74 and 84 percent during Engel et al.'s (2010) 2000–2004 sample period. We also document that, over the post-SOX period 2007–2015, the average percentage of firms that grant equity compensation to their ACs is about 96%. These statistics indicate that the use of AC equity compensation is even more pervasive. We provide detailed discussion in the section "Sample selection."

We conduct a battery of additional analyses to examine whether our main results are attributable to certain firm, AC, and auditor characteristics. We find that audit market competition at the city-level accentuates the negative relation between equity compensation and AC independence and leads to even lower audit fees. However, this negative relation disappears when (i) firms face high litigation risk, (ii) auditors have stronger bargaining power, (iii) ACs include a high proportion of accounting experts, and (iv) auditors are industry experts.

We also conduct several sensitivity tests to assess the robustness of our findings. First, we use the natural log of equity compensation (Kim et al. 2015; Sengupta and Zhang 2015) and AC chairs' equity delta and vega (which measure the sensitivity of equity holdings to changes in stock price and volatility of stock return, respectively) as alternative measures of equity compensation. We also use the natural log of cash compensation as a placebo test to examine whether it is the equity compensation and not the cash compensation that impairs AC independence. Second, we create a firm size-matched sample to test whether our results are influenced by the size effect. Third, we use a robust regression based on the MM-estimation as an alternative way to control for extreme values. Fourth, we drop firms that do not pay equity compensation to their ACs. Fifth, we restrict the sample to the postfinancial crisis period (2011–2015). Overall, the results of these robustness tests are consistent with our primary findings.

Our study makes several contributions to the auditing literature. First, although recent research uses earnings management measures (e.g., Bedard et al. 2004), restatement likelihood (e.g., Archambeault and Hermanson 2008), internal control weaknesses (e.g., Naiker and Sharma 2009), and approval for procuring nonaudit services from the current auditors (e.g., Naiker et al. 2013) to capture AC effectiveness, no prior study has examined whether equity compensation relates to AC independence, which in turn influences AC determination of audit fees. Our study provides a first attempt at addressing this question. Second, even though section 301 mandates that all AC directors be fully independent, it is possible that these directors satisfy this requirement when they are newly appointed but become less independent as they keep receiving equity compensation during their tenure. Our results suggest that increasing the responsibilities of ACs without considering their compensation is insufficient for the effective functioning of ACs. In particular, requiring ACs to be directly responsible for audit fee determination may not improve financial reporting quality as intended because AC members who receive larger equity compensation may compromise their independence. Third, DeFond and Zhang (2014) call for future auditing studies to examine the relatively less understood reasons why firms have incentives to demand audit quality. In response to this call, we show that higher equity compensation may induce AC members to compromise independence by paying lower audit fees, thereby resulting in lower demand for audit quality. Fourth, even though we find that equity compensation attenuates AC independence in determining appropriate audit fees, we also find that this negative effect disappears when auditors are industry experts with stronger bargaining power and the AC includes more accounting experts. These conditions provide regulators with insights into whether new regulations are needed to alter the balance of bargaining power between the clients and the auditors. Fifth, prior studies find that market competition is positively associated with restatement likelihood because audit firms may be forced to compete more strongly on audit fees (e.g., Numan and Willekens 2012). Our results imply that ACs that receive larger equity compensation may take advantage of this audit market competition to reduce audit fees to an even lower level. In this regard, we provide insight into how audit market competition and AC equity compensation may jointly affect AC audit fee decisions.

The remainder of this paper is organized as follows. Section 2 reviews related research and develops our hypotheses. Section 3 describes the basic research design, including the dependent and independent variable measures, econometric models, and sample selection procedures. Section 4 discusses the empirical findings and section 5 concludes the study.

2. Related literature and the relation between audit fees and AC equity compensation

Overview of related literature

Equity compensation and audit committee independence

Although section 301 of SOX requires that members of an AC be fully independent, it does not specify how AC members should be compensated to maintain their independence. Because equity compensation could induce ACs to compromise independence, AC members may have conflicting objectives that will affect their audit fee decisions. Recent empirical studies examining how equity compensation influences AC independence provide mixed evidence. On the one hand, some studies find that AC equity compensation is associated with larger discretionary accruals (e.g., Bedard et al. 2004), higher likelihood of restatement (e.g., Archambeault and Hermanson 2008), beating analyst earnings benchmarks (e.g., Campbell et al. 2015), and income misstatements (e.g., Keune and Johnstone 2015). On the other hand, Sengupta and Zhang (2015) find that larger AC equity compensation is associated with higher disclosure quality (measured by management earnings guidance and dispersion of analyst forecasts) and lower cost of equity capital.

Our study differs from these recent studies in several ways. First, although Archambeault and Hermanson (2008), Bedard et al. (2004), and Campbell et al. (2015) show that options granted to ACs are associated with lower earnings quality, they do not explore the underlying reason why AC equity compensation jeopardizes earnings quality. Our study provides a view into this "black box" by showing that equity payments may impair AC independence because equity compensation ties AC members' wealth to firms' stock prices. If equity payments make ACs less effective, they may purchase less audit effort by paying lower audit fees, which, in turn, may result in lower quality earnings. Second, management earnings guidance is only applicable to a subset of firms that make these voluntary disclosures (Chen et al. 2015). Therefore, it is unclear whether Sengupta and Zhang's (2015) findings are also valid for firms that do not make such voluntary disclosures. Third, because management earnings guidance is not subject to audit, the relation between disclosure quality and determination of audit fees is weak, at best. Fourth, although ACs are directly responsible for the quality of "firms" financial reporting, as reflected in earnings quality, they are less likely to be responsible for the quality of management guidance. Fifth, prior studies use output-based financial statement characteristics to measure the effectiveness of ACs. Unlike those studies, we examine how AC equity compensation relates to the audit fee decision (which is a direct and input-based measure of AC effectiveness). Because ACs are responsible for determining audit fees but there is little archival research on whether ACs are capable of exercising this responsibility and what factors may adversely affect the AC audit fee decision, our study provides empirical evidence that may bear important regulatory implications.

Threats to audit committee effectiveness

Naiker and Sharma (2009) find that former audit partners who are affiliated with a firm's current auditors enhance AC oversight effectiveness, which in turn improves internal controls and financial reporting quality. In a related study, Naiker et al. (2013) show that the presence of former audit partners on the ACs is negatively associated with the procurement of nonaudit services from the current auditors. Both studies conclude that the three-year cooling-off period required by the NYSE and NASDAQ is not necessary. Christensen et al. (2018) further find that affiliated partners serving on the ACs are associated with improved audit quality and a reduction in audit fees and fieldwork time, suggesting that AC affiliation affects both the quality and the efficiency of the audit. From a different perspective, Badolato et al. (2014) argue that the requirements of financial expertise and independence result in lower status (as measured by the sum of the number of directorships and number of directors graduated from elite schools) for ACs relative to management. They find that ACs with both financial expertise and high relative status exhibit lower levels of earnings management. Financial expertise alone cannot ensure AC effectiveness.

Our study differs from these three studies in two notable ways. First, whereas these studies use the presence of former audit partners affiliated with the current auditor and AC status as the threats to AC effectiveness, we examine whether equity compensation impairs or enhances the monitoring efficacy of ACs. Given the widespread use of equity to compensate AC members (Engel et al. 2010), our results inform about the implications of AC compensation practices. Second, whereas Naiker and Sharma (2009) and Naiker et al. (2013) use internal control weaknesses and purchase of nonaudit services, respectively, to capture the effectiveness of ACs, we examine how AC equity compensation relates to the audit fee decision. Although SOX was passed more than a decade ago, there is little archival research using post-SOX data on factors related to AC audit fee determination. We provide empirical evidence on this issue.

Audit committees and audit fees

More effective ACs may be associated with lower audit fees due to auditors' lower audit risk assessment or ACs' lower demand for audit effort given their own monitoring effectiveness. However, more effective ACs may also be associated with higher audit fees due to higher demand for external monitoring to protect their reputational capital and reduce their litigation risk (Carcello et al. 2002; Carcello et al. 2011). Using pre-SOX data, prior studies provide mixed evidence on the association between AC characteristics and audit fees. For example, some research finds that audit fees increase in AC percentage of outside directors' accounting or financial expertise (e.g., Carcello et al. 2002), experience, and size (e.g., Vafeas and Waegelein 2007). This result is consistent with effective ACs demanding greater audit effort. In contrast, other research reports that audit fees are negatively associated with AC expertise (e.g., Krishnan and Visvanathan 2008), which is consistent with effective ACs being associated with lower audit risk and lower demand for external monitoring. Engel et al. (2010) use both pre- and post-SOX data and find that total compensation and cash retainers paid to the ACs are positively correlated with the demand for monitoring of the financial reporting process (proxied by firm-size-deflated total audit fees).

Two recent studies use post-SOX data to investigate the association between executives' influences on ACs and audit fees. Bruynseels and Cardinaels (2014) show that ACs are less effective when CEOs have nonprofessional social ties with the AC members, leading to less demand for audit effort as captured by audit fees. In contrast, Beck and Mauldin (2014) find that, during the 2008–2009 crisis period, there is a larger audit fee reduction when the CFOs are more powerful and a smaller audit fee reduction when the ACs are more powerful.

Our study differs from this strand of literature in three important ways. First, even though Engel et al. (2010) find that the demand for monitoring of financial reporting is an important determinant of AC compensation, they do not explore whether and how equity compensation influences the effectiveness of ACs. Second, whereas prior studies test whether executives influence AC audit fee decisions, we test whether ACs that receive more equity compensation select a higher or a lower level of audit fees. Third, because CEO social ties and CFO power are not publicly observable, it is not feasible for regulators to implement rules to mitigate the adverse effects of these executive influences on AC effectiveness. By contrast, because all listed firms are mandated to disclose director compensation in their proxy statements, AC equity compensation is publicly observable to market participants. Therefore, our study provides evidence based on which regulators could potentially specify restrictions on AC compensation practices.

Relation between audit fees and AC equity compensation

According to agency theory, managers may take actions that maximize their own personal wealth rather than shareholders' wealth due to a lack of alignment in managers' and shareholders' interests (Jensen and Meckling 1976). One way that firms can mitigate this agency problem is by having a board of directors that monitors management on behalf of the shareholders. In our context, this would be an AC that specifically monitors management's financial reporting quality to reduce agency costs. Prior studies find that economic incentives may motivate ACs to focus on their own interests rather than on the interests of the other stakeholders (e.g., Dalton et al. 2003; Hillman and Dalziel 2003). Prior studies also find that when managers and AC members have incentives, they are likely to process information in a manner consistent with those incentives (e.g., Campbell et al. 2015). Overall, because principals (shareholders) delegate authority to the agents (ACs or managers) and agents' decisions affect the principals' financial benefits, agency theory deals with the conflict between principals and agents (Jensen and Meckling 1976). Therefore, we adopt the agency framework to explain whether the use of equity compensation could strengthen or weaken AC monitoring efforts on behalf of shareholders and thereby lead to stronger or weaker financial reporting oversight.

Although audit fees can be used in both demand-side and supply-side studies (DeFond and Zhang 2014), the demand side is more relevant to our study for the following reasons. First, our objective is to examine whether equity compensation influences AC audit fee decisions and the resulting earnings quality. Taking a demand-side perspective is appropriate in our setting because ACs are directly responsible for determining audit fees. Therefore, examining how auditors react to AC equity compensation will not directly address our research questions. Second, most AC studies take a demand-side perspective to examine factors that may enhance or impair AC effectiveness (e.g., Archambeault and Hermanson 2008; Beck and Mauldin 2014; Bruynseels and Cardinaels 2014; Campbell et al. 2015; Engel et al. 2010; Keune and Johnstone 2015) and audit fees are often used to test AC competencies in these demand-side studies (DeFond and Zhang 2014). Third, even though section 301 of SOX requires that ACs be directly responsible for determining audit fees to eliminate CEO/CFO influence over the auditors, these executives may offer large equity compensation to induce AC members to agree with or support management's earnings management decisions by purchasing less audit or demanding lower audit quality. Therefore, the AC agency problem is even more critical from the demand-side perspective and is especially so because investors and regulators believe that ACs can independently determine audit fees after SOX but ACs may in fact compromise independence due to high equity compensation.

Using the alignment and entrenchment theories examined in the agency literature, we reason that when AC members receive larger equity compensation from the firm, they face the following possible choices. First, following alignment theory (Jensen and Meckling 1976; Lambert et al. 2007) and prior governance literature (e.g., Fama and Jensen 1983; Monks and Minow 2011), larger equity compensation may align AC members' interests with shareholders' interests and thereby lead to more effective ACs. Because more effective oversight enhances a firm's internal governance system, ACs may demand less external monitoring from their auditors, which suggests a reduction in audit fees. We refer to this scenario as the AC *less demand* choice.³

Alternatively, more effective ACs may also demand higher audit quality in order to enhance the integrity of the firm's financial statements (Carcello et al. 2011), maintain their reputation capital (Fama 1980), and avoid litigation exposure (Naiker and Sharma 2009; Naiker et al. 2013). Prior research documents that ACs are sensitive to risk (e.g., Krishnan and Visvanathan 2008). Because ACs have a direct influence on the scope of auditors' work (Beasley et al. 2009; Carcello et al. 2002), AC demand for high-quality audits usually requires greater audit effort, which suggests an increase in audit fees. We refer to this scenario as the AC *more demand* choice.

Second, following entrenchment theory (Jensen and Ruback 1983; Shleifer and Vishny 1989), entrenched AC members may be more willing to tolerate managers' earnings manipulation to drive up stock price so that they can derive personal benefit from exercising options or selling stocks. Regulators and the financial press have expressed serious concern about whether equity compensation compromises AC independence, because stocks and options tie AC members' wealth to firms' shortterm and long-term financial performance (Millstein 2002). Prior studies indicate that economic

^{3.} By reducing audit fees, we mean that ACs still pay the normal audit fees but do not pay extra audit fees.

incentives may motivate ACs to focus on their own interests rather than on the interests of the other stakeholders (e.g., Dalton et al. 2003). If AC members' choice is consistent with self-interest, they are likely to demand less external auditing, which suggests a reduction in audit fees. We refer to this scenario as the *high incentive* choice because ACs have a high incentive to cut audit fees.

To the extent that ACs are responsible for determining audit fees and equity compensation impairs their independence, audit fee determination likely plays an important role in explaining the negative relation between AC equity compensation and earnings quality (Archambeault and Hermanson 2008; Campbell et al. 2015). Specifically, less independent ACs are likely to pay higher audit fees to induce the auditors to compromise independence through economic bonding to the firms, which may lead to lower earnings quality (Choi et al. 2010). This reasoning suggests an increase in audit fees. Because ACs have a low incentive to cut audit fees, we refer to this scenario as the *low incentive* choice.

Given that the preceding discussion indicates that the relation between audit fees and AC equity compensation cannot be ex ante predicted but can only be addressed empirically, we do not present formal hypotheses.

3. Research design

Regression model

We employ the following audit fee model (1) to examine the relation between audit fees and AC equity compensation (for brevity, we omit firm and year subscripts):

$$Ln(AF) = \alpha_0 + \alpha_1 AC_EBC\% + \alpha_2 SIZE + \alpha_3 FOREIGN + \alpha_4 SEGMENT + \alpha_5 CURRENT + \alpha_6 INVREC + \alpha_7 ROA + \alpha_8 LOSS + \alpha_9 MB + \alpha_{10} LEV + \alpha_{11} CFO + \alpha_{12} SPECIAL + \alpha_{13} RD + \alpha_{14} RESTRUCTURE + \alpha_{15} MA + \alpha_{16} RET_VOL + \alpha_{17} ICW + \alpha_{18} CEO_DELTA + \alpha_{19} CEO_VEGA + \alpha_{20} CFO_DELTA + \alpha_{21} CFO_VEGA + \alpha_{22} CEO_CHANGE + \alpha_{23} BDSIZE + \alpha_{24} BDIND + \alpha_{25} ACSIZE + \alpha_{26} TENURE + \alpha_{27} AUDIT_CHANGE + \alpha_{28} BIG4 + \alpha_{29} LOG (NAS) + Industry and Year FE + \varepsilon. (1)$$

We provide detailed variable definitions in the Appendix.

Model (1) relates our primary dependent variable, the natural log of audit fees (Ln(AF)), to AC equity compensation (AC_EBC%). We follow Sengupta and Zhang (2015) and measure AC_EBC% as the average of the equity compensation ratios across all AC members of a firm. We control for both demand-side and supply-side audit fee determinants in model (1), using control variables identified in prior research (e.g., Beardsley et al. 2019; Chen et al. 2015; Engel et al. 2010; Kim et al. 2015). In addition, because we focus on the demand side to develop the relation between audit fees and AC equity compensation, we include more supply-side determinants to strengthen the reliability of our results. From the supply-side perspective, we consider five major types of firm-specific characteristics: operational complexity (SIZE, FOREIGN, and SEGMENT); operating risks such as liquidity (CURRENT), profitability (ROA and LOSS), inherent risk (INVREC), and future growth opportunity (MB); financial risks such as leverage (LEV), cash flow from operations (CFO), special items reported in the income statement (SPECIAL), R&D expenditure (RD), debt restructuring (RESTRUCTURE), merger and acquisition (MA), and stock return volatility (RET_VOL); control risk (ICW); governance risks such as CEO and CFO equity incentives (CEO_DELTA, CEO_VEGA, CFO_DELTA, and CFO_VEGA) and CEO turnover (CEO_CHANGE). From the demand-side perspective, we control for board characteristics such as board size (BDSIZE), board independence (BDIND), and AC size (ACSIZE). We also include auditor characteristics such as auditor tenure (TENURE), auditor turnover (AUDIT_ CHANGE), auditor quality (BIG4), and nonaudit service fees (LOG(NAS)). We provide detailed variable definitions in the Appendix.

TABLE 1

Analyses of AC equity compensation

Panel A: Distribution by year

	Firms	not offering equity compensation	Firm	ns offering equity compensation	Number obs	rs of firm-year servations
Year	Number	Percentage of observations in each year (%)	Number	Percentage of observations in each year (%)	Number	Percentage of total (%)
2007	26	6.44	378	93.56	404	10.96
2008	22	5.02	416	94.98	438	11.89
2009	19	4.55	399	95.45	418	11.34
2010	20	4.76	400	95.24	420	11.40
2011	13	2.95	427	97.05	440	11.94
2012	12	2.85	409	97.15	421	11.42
2013	14	3.25	417	96.75	431	11.70
2014	21	5.22	381	94.78	402	10.91
2015	7	2.25	304	97.75	311	8.44
Total	154	4.18	3,531	95.82	3,685	100.00

Panel B: Distribution by industry

Industry (SIC Code)	Number	Percentage of total observations
Agriculture (0100–0999)	15	0.41
Mining and construction (1000–1999)	82	2.23
Food (2000–2111)	183	4.97
Textiles and printing/Publishing (2200–2799)	178	4.83
Chemicals (2800–2824, 2840–2899)	148	4.02
Pharmaceuticals (2830–2836)	160	4.34
Extractive (1300–1399, 2900–2999)	269	7.30
Durable manufacturing (3000–3999, excluding 3570–3579 and 3670–3679)	782	21.22
Transportation (4000–4899)	231	6.27
Utilities (4900–4999)	311	8.44
Retailing (5000–5999)	512	13.89
Services (7000–9999, excluding 7370–7379)	231	6.27
Computers (3570–3579, 3670–3679, 7370–7379)	583	15.82
Total	3,685	100.00

Panel C: Distribution by type of AC compensation

	Average an	mount of annual com	pensation (\$)	Compensation s	structure (%)
Year	Cash	Equity	Total	Cash	Equity
2007	81,844.68	187,215.65	272,023.14	43.31	56.73
2008	76,102.49	144,203.79	221,032.59	45.16	54.86
2009	78,856.17	174,925.09	254,047.08	39.96	60.06
2010	90,497.23	179,033.45	270,370.54	41.16	58.92
2011	87,514.37	183,983.96	272,049.80	40.61	59.44
2012	90,116.63	167,002.12	257,059.29	40.55	59.48

(The table is continued on the next page.)

Panel C: Di	stribution by type	of AC compensation			
	Average a	nount of annual com	pensation (\$)	Compensation s	structure (%)
Year	Cash	Equity	Total	Cash	Equity
2013	86,378.62	205,042.67	291,263.85	39.78	60.26
2014	88,353.08	204,217.49	292,520.45	40.11	59.91
2015	95,283.15	211,072.59	306,293.28	41.31	58.71
Average	86,105.16	184,077.42	270,740.00	41.33	58.71

TABLE 1 (continued)

Notes: Figures reported in panel C represent the dollar amounts and ratios of the average cash and equity compensation a firm pays to *all* its AC members in each year. The cash compensation does not include meeting fees.

In addition to the variables listed above, we include year and industry fixed effects to control for the potential effects of macroeconomic, regulatory, and industry-specific factors on audit fees (Adams and Ferreria 2009) and report *t*-statistics that are corrected for heteroskedasticity.

Sample selection

The initial sample consists of 9,374 firm-year observations with complete board characteristics and compensation data in BoardEx for the years 2007 through 2015. We exclude 1,893 observations representing firms in the financial services industry because firms from this industry have different financial reporting characteristics. We lose 663, 198, and 172 observations when we merge this sample with Audit Analytics, Compustat, and CRSP, respectively. We also lose 1,829 observations with missing data in ExecuComp when we compute CEO's and CFO's delta and vega. We truncate observations that fall in the top and bottom 1% of the sample distributions of the dependent and all continuous independent variables, resulting in a final sample of 742 firms with 3,685 firm-year observations.

Table 1 reports the distributions of our sample. Panel A shows that the yearly percentage of firms that do not offer equity compensation to their ACs decreased from a high of 6.44% in 2007 to a low of 2.25% in 2015. This result is consistent with the current trend of widespread use of stocks and options to compensate firms' outside directors. Over the sample period, about 96% of the observations offer equity compensation. The last two columns of panel A indicate that the sample is evenly distributed over 2007–2015, except for the last year when there are fewer observations.⁴ Panel B shows that Durable manufacturing (21.22%), Computer (15.82%), Retailing (13.89%), Utilities (8.44%), Extractive (7.30%), and Services (6.27%) account for about 73% of the sample.

Table 1, panel C reports the average amounts and proportions of total cash and equity compensation paid to the ACs in our sample. BoardEx provides only the total amount of equity compensation and does not separate meeting fees but includes them in "Others" and "Total Compensation." Therefore, the total cash compensation reported in panel C does not include meeting fees and we do not decompose equity compensation into stock and option components. Panel C indicates that the average cash compensation increases while the cash proportion

^{4.} We checked the raw compensation data and found that, as compared to 2007–2014, year 2015 has the smallest number of observations in BoardEx.

TABLE 2			
Descriptive	statistics	(N =	3,673)

Variables	Min	Q1	Mean	Median	Q3	Max	SD
Ln(AF)	12.795	14.402	15.051	15.076	15.672	17.328	0.910
AC_EBC%	0.000	0.486	0.587	0.596	0.723	1.000	0.207
SIZE	4.736	7.856	8.708	8.796	9.708	12.252	1.419
FOREIGN	0.000	0.000	0.376	0.000	1.000	1.000	0.485
SEGMENT	0.000	4.000	6.362	6.000	9.000	22.000	3.723
CURRENT	0.476	1.180	2.005	1.686	2.485	7.858	1.188
INVREC	0.020	0.095	0.212	0.194	0.295	0.680	0.137
ROA	-0.227	0.062	0.106	0.100	0.146	0.371	0.072
LOSS	0.000	0.000	0.118	0.000	0.000	1.000	0.322
MB	-18.834	1.585	3.194	2.483	3.952	29.405	3.270
LEV	0.000	0.126	0.241	0.238	0.343	0.829	0.158
CFO	-0.025	0.019	0.030	0.028	0.039	0.093	0.016
SPECIAL	0.000	1.000	0.823	1.000	1.000	1.000	0.381
RD	0.000	0.000	0.503	1.000	1.000	1.000	0.500
RESTRUCTURE	0.000	0.000	0.478	0.000	1.000	1.000	0.500
MA	0.000	0.000	0.536	1.000	1.000	1.000	0.499
RET_VOL	0.026	0.039	0.046	0.047	0.054	0.061	0.010
ICW	0.000	0.000	0.017	0.000	0.000	1.000	0.129
CEO_DELTA	4.553	139.159	773.917	349.864	794.828	17343.530	1530.470
CEO_VEGA	0.000	22.697	181.997	104.068	257.423	1405.400	218.795
CFO_DELTA	0.478	27.467	113.541	65.213	145.019	983.069	133.471
CFO_VEGA	0.000	5.299	44.670	25.277	61.980	404.217	55.357
CEO_CHANGE	0.000	0.000	0.104	0.000	0.000	1.000	0.305
BDSIZE	4.000	8.000	9.473	9.000	11.000	16.000	2.509
BDIND	0.600	0.889	0.935	1.000	1.000	1.000	0.086
ACSIZE	3.000	4.000	4.554	4.000	5.000	8.000	1.185
TENURE	1.000	3.000	7.690	8.000	16.000	32.000	8.004
AUDIT_CHANGE	0.000	0.000	0.018	0.000	0.000	1.000	0.135
BIG4	0.000	1.000	0.964	1.000	1.000	1.000	0.185
LOG(NAS)	0.000	12.181	12.871	13.329	14.268	16.645	2.542

Notes: This table presents the descriptive statistics of all variables used in our analyses. See Appendix for the definitions of variables. For ease of readability, we report the original means and medians of *SEGMENT*, *BDSIZE*, *ACSIZE* and *TENURE*. In the regressions, we use the natural log values of these numbers.

decreases during 2007–2014 and then increases in 2015. In contrast, the average equity compensation increases and the equity proportion remains relatively stable at around 60% between 2009 and 2014 and decreases slightly in 2015. Lastly, the average total compensation paid in 2015 is 38.57% larger than the amount in 2008. This large increase implies that the demand for AC monitoring increased during our sample period, consistent with recent survey reports (e.g., National Association of Corporate Directors (NACD) 2015).

Table 2 reports the descriptive statistics for all variables used in the study. The mean of Ln (*AF*) is 15.051, which is comparable to the mean of 14.095 reported in Kim et al. (2015). On average, the ratio of equity compensation to total compensation across all AC members in a firm is 58.7%. We note that firms have 93.5% independent directors serving on their boards and only 1.7% of the firms have internal control weaknesses. In addition, almost 97% of the firms are audited by a Big 4 auditor. These statistics suggest that our sample firms have good internal and

Pearsc	n correlati	on coefficie	ents												
	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)	(6)	(10)	(11)	(12)	(13)	(14)	(15)
(1)	1.000														
(5)	0.035	1.000													
3)	0.780	0.048	1.000												
4	0.108	0.020	-0.084	1.000											
(2)	0.373	-0.033	0.161	0.263	1.000										
9	-0.300	0.024	-0.406	0.152	0.091	1.000									
6	-0.022	-0.063	-0.228	0.062	0.089	0.143	1.000								
(8)	-0.016	0.107	-0.004	0.010	-0.043	0.113	0.217	1.000							
6	-0.077	-0.099	-0.170	0.043	-0.021	0.026	-0.017	-0.482	1.000						
(10)	0.022	0.129	-0.001	-0.010	-0.034	0.013	0.046	0.350	-0.170	1.000					
(11)	0.166	-0.041	0.262	-0.108	-0.079	-0.327	-0.195	-0.160	0.104	0.028	1.000				
(12)	-0.159	0.147	-0.086	0.002	-0.107	0.096	-0.037	0.679	-0.314	0.268	-0.263	1.000			
(13)	0.254	0.012	0.108	0.101	0.161	-0.049	-0.072	-0.189	0.143	-0.040	0.157	-0.206	1.000		
(14)	0.229	0.096	-0.076	0.239	0.339	0.265	060.0	0.032	0.010	0.149	-0.182	-0.006	0.212	1.000	
(15)	0.276	0.022	0.070	0.169	0.222	0.00	0.046	-0.126	0.129	0.000	0.085	-0.158	0.441	0.351	1.000
(16)	0.162	0.045	0.014	0.099	0.170	0.032	0.071	0.030	-0.027	0.011	-0.036	-0.022	0.193	0.204	0.142
(17)	-0.079	0.042	-0.035	0.018	-0.030	0.066	0.000	0.027	-0.037	-0.048	-0.057	0.015	-0.001	-0.012	-0.025
(18)	-0.001	-0.048	-0.075	0.025	0.016	0.003	0.018	-0.082	0.096	-0.019	0.016	-0.035	0.033	0.033	0.027
(19)	0.099	0.132	0.128	-0.001	-0.055	-0.004	-0.016	0.146	-0.097	0.109	-0.013	0.140	-0.026	-0.001	-0.040
(20)	0.426	0.103	0.410	0.047	0.139	-0.055	-0.029	0.167	-0.143	0.114	-0.008	0.096	0.070	0.182	0.106
(21)	0.258	0.184	0.306	-0.017	0.011	-0.052	0.020	0.222	-0.168	0.198	-0.035	0.174	-0.025	0.071	-0.025
(22)	0.348	0.127	0.355	0.027	0.077	-0.019	-0.002	0.162	-0.134	0.102	-0.013	0.099	0.052	0.162	0.083
(23)	0.039	-0.016	0.014	0.024	0.023	-00.00	0.010	-0.053	0.066	-0.013	-0.002	-0.037	0.038	0.029	0.065
(54)	0.552	-0.019	0.605	-0.045	0.138	-0.321	-0.099	-0.010	-0.112	-0.005	0.223	-0.127	0.142	-0.015	0.123
(25)	-0.047	0.099	-0.102	0.019	-0.001	0.081	0.004	0.033	0.028	0.027	-0.058	0.066	0.005	0.107	0.049
(36)	0.324	-0.057	0.348	-0.079	0.084	-0.221	0.012	-0.017	-0.041	0.008	0.154	-0.106	0.073	-0.005	0.055
(27)	0.245	0.015	0.280	-0.043	0.070	-0.087	0.030	0.091	-0.096	0.076	0.044	0.024	0.003	0.054	0.061
(28)	-0.053	-0.030	-0.065	0.043	-0.006	0.026	0.000	-0.056	0.063	-0.026	0.003	-0.049	0.027	-0.005	-0.002
(29)	0.225	0.059	0.257	0.004	0.061	-0.124	-0.112	0.045	-0.048	0.041	0.125	0.033	0.076	0.026	0.063
(30)	0.550	0.052	0.459	0.081	0.200	-0.150	-0.019	0.033	-0.097	0.034	0.087	-0.084	0.160	0.192	0.203
												The table is	continued	on the nex	t page.)

TABLE 3

TABI	LE 3 (contir	nued)													
	(16)	(17)	(18)	(19)	(20)	(21)	(22)	(23)	(24)	(25)	(26)	(27)	(28)	(29)	(30)
(16)	1.000														
(17)	-0.022	1.000													
(18)	0.012	-0.040	1.000												
(19)	0.034	-0.016	-0.029	1.000											
(20)	0.074	0.063	-0.025	0.286	1.000										
(21)	0.045	-0.007	-0.051	0.266	0.384	1.000									
(22)	0.076	0.084	-0.048	0.182	0.718	0.625	1.000								
(23)	-0.025	-0.006	0.018	-0.107	-0.111	-0.040	-0.008	1.000							
(54)	0.002	-0.029	-0.036	0.001	0.277	0.139	0.218	0.033	1.000						
(25)	0.017	0.045	0.021	-0.009	-0.022	-0.049	-0.051	-0.041	-0.277	1.000					
(36)	-0.010	-0.020	-0.010	-0.024	0.110	0.023	0.054	0.010	0.513	0.055	1.000				
(27)	-0.018	0.035	-0.025	0.043	0.175	0.115	0.152	0.015	0.280	-0.021	0.212	1.000			
(28)	0.047	-0.074	0.029	-0.020	-0.039	-0.017	-0.043	-0.020	-0.045	-0.001	-0.029	-0.150	1.000		
(29)	0.045	0.010	0.002	0.059	0.112	0.092	0.106	0.012	0.220	-0.016	0.101	0.190	-0.050	1.000	
(30)	0.104	-0.022	-0.011	0.071	0.254	0.163	0.230	0.016	0.375	-0.070	0.186	0.198	-0.056	0.158	1.000
Notes	• In this tab	I = I = I	$(AF) \cdot (2) =$	AC ERCW	IZIS = (2)	$F_{1}(4) = FC$	REIGN (5	= SEGMF	ENT. (6) = 1	CURRENT	IANI = (L)	$REC \cdot (8) = 1$	ROA· (9) =	-5501	
1000	· •••• ••••		- (-) () 												
(10) =	= <i>MB</i> ; (11) :	= <i>LEV</i> ; (12)	= CFO; (1)	(3) = SPECI	AL; (14) =	<i>RD</i> ; (15) =	RESTRUC	TURE; (16	M = MA; (1)	$7) = RET_V$	/OL; (18) =	: ICW; (19)	$= CEO_DH$	ELTA;	
(20) =	: CEO_VEC	3A; (21) = (CF0_DELI	A; (22) = C	FO_VEGA	(23) = CE	CHANC	iE; (24) = I	BDSIZE; (2	5) = BDIN	D; (26) = A	CSIZE; (27) = TENUR	Ε;	
(28) =	AUDIT_C	THANGE; (2	(9) = BIG4;	(30) = LO(G(NAS). See	e Appendix	for variabl	le definition	ns. Correlati	ion coeffici	ents that are	e significant	at the 0.10) level or t	etter
are in	bold.														

external governance. Lastly, a board has about 10 directors and an AC has about 5 independent directors.

Table 3 presents Pearson correlation coefficients between the dependent and the independent variables. First, the correlation between Ln(AF) and $AC_EBC\%$ is positive and significant. Because this correlation is pairwise, the coefficient sign may differ in the multivariate analysis that controls for relevant differences. Second, Ln(AF) is negatively related to *CURRENT* and *CFO* (significant at the 0.05 level or better), implying that auditors charge lower audit fees if clients are less risky due to their high liquidity and cash flow from operations. Third, Ln(AF) is positively related to *SIZE*, *FOREIGN*, *SEGMENT*, *LEV*, *RESTRUCTURE*, *MA*, *CEO_VEGA*, and *CFO_VEGA* (significant at the 0.05 level or better), which indicates that auditors charge higher audit fees if firms are riskier due to higher complexity, operating losses, higher financial leverage, debt restructuring, mergers, and CEO/CFO equity incentives. Fourth, Ln(AF) is positively associated with *BDSIZE* and *ACSIZE* (significant at the 0.01 level), indicating that larger boards and ACs are likely to pay higher audit fees. These results show that board governance is positively related to audit fees.⁵

4. Empirical results

Main results

Relation between audit fees and AC equity compensation

We report the estimation results of model (1) in Table 4. The adjusted R^2 is 0.823, indicating that the model explains a large portion of the variation in audit fees. The coefficient on $AC_EBC\%$ is negative and significant at the 0.05 level, implying that the higher the proportion of AC equity compensation, the lower the audit fees. Using the mean value of Ln(AF) as the reference point (the raw value is exp(15.051) = \$3,440,062), this coefficient indicates that a one-unit increase in $AC_EBC\%$ is associated with a reduction of \$297,710 in raw audit fees.^{6,7} These results are consistent with the *less demand* choice scenario and also with the *high incentive* choice scenario. We attempt to distinguish between these two scenarios in the following section.

Whereas we find a negative association between audit fees and AC equity compensation, recent studies report a positive association between audit fees and CEO/CFO equity incentives (e.g., Billings et al. 2014; Chen et al. 2015; Kim et al. 2015). Three possible reasons may explain this difference. First, prior audit fee studies usually adopt a supply-side perspective to examine whether auditors are able to incorporate certain risks related to management compensation into their fee determination. Because CEO and CFO equity incentives are regarded as sources of risk (i.e., equity vega induces risk-taking behavior), auditors will charge

^{5.} The univariate correlations between Ln(AF) and LOSS and BDIND are negative, suggesting that lower audit fees are associated with firms incurring operating losses or with more independent directors on the boards. In addition, the correlation between Ln(AF) and ICW is not significant. However, the regression results of our model (1) show that the coefficients on LOSS, ICW, and BDIND are all positive and significant at the 0.05 level or better. These results suggest that after controlling for correlated variables that affect audit fees, LOSS, ICW, and BDIND are positively associated with audit fees, consistent with prior studies.

^{6.} We find that a one standard deviation (0.207, see Table 2) increase in $AC_EBC\%$ is associated with an audit fee reduction of \$59,104 (i.e., $[exp(0.083 \times 0.207) - 1] \times $3,440,062$). This amount is important because it is almost equal to the highest salary level the Big 4 pay to an entry-level staff (Corporate Finance Institute 2020). In response to such audit fee reduction, an audit firm may choose to reduce the recruitment of new audit staff or assign less experienced auditors to each audit engagement. These human resources policies will harm audit firms' long-term audit quality because their human capital will decrease.

^{7.} Because the audit fees are log-transformed, it may not be necessary to delete observations in the top and bottom 1% of Ln(AF). We therefore reestimate model (1) after including the additional 29 observations. Untabulated results show that the coefficient on $AC_EBC\%$ is negative (i.e., -0.066) and significant at the 0.10 level (p < 0.059), which represents an audit fee reduction of \$46,998 (i.e., $-0.066 \times 0.207 \times \exp(15.051)$). This robustness check indicates that our main results reported in Table 4 are not sensitive to truncating observations for Ln(AF).

754 Contemporary Accounting Research

TABLE 4

	Dep.	= Ln(AF)
Variables	Coeff.	<i>t</i> -stat.
Intercept	9.166	48.26***
AC_EBC%	-0.083	-2.61***
SIZE	0.482	55.68***
FOREIGN	0.078	5.22***
SEGMENT	0.120	8.31***
CURRENT	-0.045	-6.22***
INVREC	1.233	14.36***
ROA	0.326	1.83*
LOSS	0.117	4.77***
MB	-0.002	-0.73
LEV	0.005	0.08
CFO	-3.484	-5.07***
SPECIAL	0.116	5.68***
RD	0.155	6.33***
RESTRUCTURE	0.058	3.59***
MA	0.037	2.62***
RET_VOL	-1.299	-0.60
ICW	0.228	3.68***
CEO_DELTA	0.000	-0.02
CEO_VEGA	0.000	6.21***
CFO_DELTA	0.000	1.15
CFO_VEGA	0.000	-1.31
CEO_CHANGE	0.045	2.12**
BDSIZE	0.235	6.11***
BDIND	0.411	4.81**
ACSIZE	-0.014	-0.44
TENURE	-0.015	-1.87*
AUDIT_CHANGE	-0.059	-1.11
BIG4	0.016	0.37
LOG(NAS)	0.031	8.28***
Industry and Year FE	Inc	cluded
Adi. R^2	C	.821

Relation between	audit fees	and AC	equity-based	compensation	(N = 3.685)
	addit rees		equity bused	compensation	(1) = 3,000)

Notes: This table presents the results of estimating the relation between audit fees and AC equity compensation. The dependent variable is the natural log of total audit fees. See Appendix for variable definitions. All the *t*-statistics are corrected for heteroskedasticity. *, **, and *** denote two-tailed significance levels at 10%, 5%, and 1%, respectively.

higher audit fees. In contrast, we take a demand-side approach to examine whether AC independence is affected by equity compensation which, in turn, influences AC audit fee decisions. Second, we estimate a lagged change model in which we regress the change in Ln(AF) (from t - 1 to t) on the change in $AC_EBC\%$ (from t - 2 to t - 1), the change in stock returns (from t - 1 to t), and the changes in all other control variables (from t - 1 to t) in model (1). The coefficient on the change in $AC_EBC\%$ is 0.018, which is not significantly different from zero (t = 1.01). This result suggests that auditors do not regard the change in AC equity compensation (from t - 2 to t - 1) as a risk source and react by increasing audit fees (from t - 1 to t). Therefore, the supply-side story may not be as appropriate for our study. Third, we re-estimate model (1) after

removing AC_EBC%. The results show that the coefficients on CEO_VEGA and CFO_VEGA are both positive and significant at the 0.05 level or better (t = 5.30, p < 0.001 and t = 2.06, p < 0.039, respectively), consistent with prior supply-side audit fee studies.⁸

Turning to the control variables, the signs of the coefficients are generally consistent with expectations. Audit fees are positively associated with operational complexity (*SIZE*, *FOREIGN*, and *SEGMENT*), inherent risk (*INVREC*), operating risk (*LOSS*), financial risk (*SPECIAL*, *RD*, *RESTRUCTURE*, and *MA*), control risk (*ICW*), audit risk (*CEO_VEGA* and *CEO_CHANGE*), board governance (*BDSIZE* and *BDIND*), and nonaudit service fees (LOG(NAS))⁹ (all are significant at the 0.10 level or better). These results imply that auditors are able to adjust audit fees upward in response to clients' risks, consistent with prior studies (e.g., Beardsley et al. 2019; Hogan and Wilkins 2008; Kim et al. 2015). In contrast, audit fees are negatively related to liquidity (*CURRENT*) and cash flow from operations (*CFO*) (significant at the 0.01 level), suggesting that auditors regard firms with higher liquidity or cash flow from operations as less risky and, therefore, charge lower audit fees.¹⁰

Distinguishing between the alignment and the entrenchment explanations

The results reported in Table 4 are consistent with both the *less demand* and the *high incentive* scenarios but do not distinguish which scenario better describes the observed negative relation between audit fees and AC equity compensation. We attempt to do so with the following tests. Under the *less demand* scenario, equity compensation should induce ACs to monitor firms' financial reporting more closely, which suggests a positive relation between earnings quality and AC equity compensation. In contrast, under the *high incentive* scenario, we should observe a negative relation between earnings quality and AC equity compensation. To distinguish between these two scenarios, we regress earnings quality (*EQ*) proxies on $AC_EBC\%$ and a set of controls using the following model (2):

 $EQ = \alpha_0 + \alpha_1 AC_EBC\% + \alpha_2 SIZE + \alpha_3 SEGMENT + \alpha_4 CURRENT + \alpha_5 INVREC$ $+ \alpha_6 ROA + \alpha_7 LOSS + \alpha_8 LEVA + \alpha_9 RD + \alpha_{10} SALE_VOL + \alpha_{11} RET_VOL + \alpha_{12} ICW$ $+ \alpha_{13} CEO_CHANGE + \alpha_{14} BDSIZE + \alpha_{15} BDIND + \alpha_{16} ACSIZE + \alpha_{17} TENURE$ $+ \alpha_{18} AUDIT_CHANGE + \alpha_{19} BIG4 + \alpha_{20} LOGMV + Industry and Year FE + \varepsilon.$ (2)

We provide detailed variable definitions in the Appendix.

^{8.} Even though section 301 of SOX requires that ACs be directly responsible for determining audit fees to eliminate CEO/CFO's influence over the auditors, AC members and CEOs/CFOs face similar agency conflicts for two reasons. First, prior studies find that economic incentives may motivate ACs to focus on their own interests rather than on the interests of the other stakeholders (e.g., Dalton et al. 2003; Hillman and Dalziel 2003). Prior studies also find that when managers and AC members have incentives, they are likely to process information in a manner consistent with those incentives (e.g., Campbell et al. 2015). Second, as indicated in the section "Relation between audit fees and AC equity compensation," although CEOs/CFOs cannot determine the audit fees, they may offer large equity compensation to induce the AC members to agree with or support management's earnings management decisions by purchasing less audit or demanding lower audit quality. This is the key issue we want to examine in our study.

^{9.} As a robustness check, we reestimate model (1) after replacing *LOG(NAS)* with the ratio of NAS fees to total audit fees. Our results reported in Table 4 remain unchanged using this alternative measure of NAS fees.

^{10.} Different from prior audit fee studies, the coefficients on *BIG4* are not significant in the main test and most of the additional and robustness tests. We provide two explanations for this result. First, we use our final sample to estimate an auditor choice model and find that equity compensation does not influence AC members' Big 4 auditor choice decisions. This finding and the results reported in Table 4 imply that, even though ACs that receive large equity compensation do not want to buy more audit effort from the Big 4 auditors, they select Big 4 auditors as a signal of financial reporting quality to firms' stakeholders. Second, we use the annual medians of *AC_EBC*% to separate our sample into high and low *AC_EBC*% subsamples and reestimate model (1) after excluding *AC_EBC*%. Untabulated results show that the coefficient on *BIG4* is not significant in the high *AC_EBC*% subsample but is significant in the low *AC_EBC*% subsample. These results provide demand-side evidence that ACs appear not to pay a fee premium to the Big 4 when they receive large equity compensation.

In model (2), we use three measures to capture firms' earnings quality: performance-adjusted discretionary accruals (absolute value *Abs_DA* and signed values *Pos_DA* and *Neg_DA*), restatement likelihood (*RESTATE*), and real earnings management (*REM1* and *REM2*). We provide detailed definitions of these measures in the Appendix.

Table 5 reports that the coefficients on *AC_EBC*% are significantly positive at the 0.05 level when the dependent variables are *Abs_DA*, *Pos_DA*, *RESTATE*, *REM1*, and *REM2*, suggesting that earnings quality decreases as the proportion of AC equity compensation increases. These results are consistent with the *high incentive* scenario, which indicates that the negative relation between audit fees and AC equity compensation likely decreases the auditors' probability of detecting earnings management.¹¹ Our results are also consistent with prior studies that suggest that auditors could decrease audit hours or use less experienced auditors in response to reduced audit fees, which would decrease the likelihood of detecting firms' earnings management (e.g., Beck and Mauldin 2014; PCAOB 2010a, 2010b, 2010c).

Prior research finds that managers may use accrual-based and real earnings management as substitutes (e.g., Zang 2012; Cohen et al. 2008). Based on these findings, Chi et al. (2011) use audit fees to proxy for audit quality and find a positive (negative) association between audit fees and real (accrual-based) earnings management. They interpret their results as evidence that because higher audit quality constrains accrual-based earnings management, firms resort to more costly real earnings management to accomplish their earnings management objectives. However, our Table 5 results imply that audit fees are negatively related to both accrual-based and real earnings management.¹² This apparent inconsistency may be explained by the following two possible reasons. First, because Chi et al. (2011) focus on firms that have strong incentives to manage earnings upward, they restrict their sample to firms that meet or just beat earnings benchmarks or issue seasoned equity offerings. Unlike Chi et al. (2011), we place no restrictions on our sample. This difference in sample composition is one possible reason for the apparent inconsistency in the observed relation between audit fees and real earnings management between Chi et al. (2011) and our study. Second, Cohen et al. (2008) and Zang (2012) use both pre- and post-SOX samples (i.e., 1987–2005 and 1987–2008, respectively) to test firms' trade-off between accrual-based and real earnings management. In contrast, we use a more recent, post-SOX only sample (i.e., 2007–2015). It is possible that firms' trade-off strategy has changed in recent years. To provide *indirect* evidence for this conjecture, we follow Beck and Mauldin (2014) and use AC chair tenure and CFO tenure to control for the relative power of the AC chair and the CFO in determining audit fees.¹³ Whereas Beck and Mauldin (2014) report a mean of -0.154 for AC chair relative power (which suggests that on average the CFOs are relatively more powerful than the AC chairs) in their 2006–2009 sample, we find that the mean of AC chair relative power is 0.036 (which indicates that on average the AC chairs are relatively more powerful than the CFOs) in our 2007–2015 sample. To better understand the difference in relative power between Beck and Mauldin (2014) and our study, we compute the relative power for three subperiods of our sample: 2007–2009, 2010–2012, and 2013–2015. We find

^{11.} Zang (2012) finds that firms can adjust the extent of accrual-based earnings management based on the observed impact of real earnings management. Although we include Abs_DA because firms may potentially substitute accrual-based and real earnings management, as a robustness check, we reestimate our two *REM* models after dropping Abs_DA . The empirical results show that the coefficients on $AC_EBC\%$ (i.e., 0.788 and 0.786 in the *REM1* and *REM2* models, respectively) remain positive and significant at the 0.05 level (t = 2.45, p < 0.015 and t = 2.47, p < 0.014, respectively), which indicates that our *REM* results are not affected by the inclusion of Abs_DA .

^{12.} We thank an anonymous reviewer for this suggestion regarding prior literature on the substitution of accruals-based and real earnings-based management and the association with audit quality.

^{13.} We sort all firm-year observations into quartiles based on AC chair tenure (*AC_QUARTILE*) and CFO tenure (*CFO_QUARTILE*). Each set of quartile values ranges from 1 to 4. We then create a variable *RELATIVE_AC_CFO* (which equals *AC_QUARTILE* – *CFO_QUARTILE*) to capture the relative power of the AC chair and the CFO. By definition, *RELATIVE_AC_CFO* ranges from -3 to +3. Positive values of *RELATIVE_AC_CFO* indicate higher AC power relative to CFO power, and vice versa. Therefore, we classify firms with positive and negative *RELATIVE_AC_CFO* as having high and low AC relative power, respectively.

	Dep. =	Abs_DA	Dep. =	Pos_DA	Dep. =	Neg_DA	Dep. = 1	RESTATE	Dep. =	REMI	Dep. =	REM2
Variables	Coeff.	t-stat.	Coeff.	t-stat.	Coeff.	t-stat.	Coeff.	t-stat.	Coeff.	t-stat.	Coeff.	t-stat.
Intercept	0.018	0.95	0.065	2.27**	0.010	0.51	0.764	0.60	1.036	0.74	1.223	06.0
$AC_{EBC\%}$	0.009	2.08^{**}	0.017	2.35**	-0.002	-0.40	0.597	1.96^{**}	0.767	2.19^{**}	0.776	2.22**
SIZE	-0.002	-1.40	-0.003	-1.27	0.001	1.06	-0.157	-2.44**	-0.027	-0.37	-0.066	-0.92
SEGMENT	0.001	0.33	-0.001	-0.24	-0.002	-1.13	0.130	1.04	0.053	0.57	0.054	0.59
CURRENT	0.003	2.39**	0.002	1.09	-0.001	-1.06	-0.103	-1.60	-0.092	-1.19	-0.091	-1.18
INVREC	-0.018	-1.16	0.020	0.80	0.041	2.41^{**}	-0.386	-0.60	0.921	1.05	0.582	0.67
ROA	-0.048	-1.59	0.160	3.85***	0.231	6.35***	-3.892	-2.83***	-1.104	-1.15	-0.106	-0.11
SSOT	0.026	7.53***	-0.008	-1.82	-0.036	-8.15^{***}	-0.211	-1.06	-0.003	-0.03	0.045	0.40
LEV	0.009	1.18	-0.006	-0.62	-0.032	-3.33***	1.122	2.66^{***}	-0.008	-0.02	0.007	0.02
CFO	1.351	8.77***	-1.567	-4.52***	-2.907	-18.71^{***}	-2.388	-0.44	-1.359	-0.23	-4.258	-0.73
RD	0.007	2.46**	0.006	1.43	-0.005	-1.22	-0.072	-0.36	-0.226	-1.68*	-0.196	-1.50
SALE_VOL	0.027	2.38**	0.009	0.98	-0.035	-2.37**	0.539	1.56	0.170	0.48	-0.040	-0.12
RET_VOL	-0.488	-1.45	-0.657	-1.22	0.391	1.13	10.974	0.59	-18.819	-2.18 **	-15.745	-1.88*
ICW	0.000	-0.01	0.014	1.42	-0.001	-0.09	2.981	9.48***	0.061	0.35	0.072	0.44
CE0_CHANGE	0.003	1.03	0.002	0.61	-0.002	-0.66	-0.035	-0.19	-0.225	-1.60	-0.251	-1.83*
BDSIZE	0.001	0.26	0.005	0.53	-0.003	-0.62	-0.343	-1.10	-0.292	-0.77	-0.233	-0.62
BDIND	-0.004	-0.33	-0.026	-1.38	-0.007	-0.65	-2.273	-3.33***	-0.605	-0.80	-0.695	-0.93
ACSIZE	0.008	1.75*	0.003	0.46	-0.007	-1.32	0.328	1.19	0.571	1.61	0.536	1.52
TENURE	0.000	0.05	0.000	0.03	0.000	0.21	0.015	0.21	-0.023	-0.31	-0.025	-0.34
AUDIT_CHANGE	0.004	0.64	0.012	0.94	-0.001	-0.21	0.250	0.63	0.142	0.77	0.104	0.59
BIG4	-0.001	-0.11	-0.001	-0.18	-0.002	-0.34	0.793	1.01	0.532	1.41	0.548	1.46
Abs_DA									1.006	1.22	0.984	1.22
Industry and Year FE	Inci	luded	Incl	luded	Incl	nded	Incl	nded	Inclu	ıded	Inclu	ded
N	З,	629	1,	249	5,2	380	ų	568	2,7	69	2,7	59
Adj. R^2 /Pseudo R^2	0.	313	0.	180	0.	486	0.	124	0.0	81	0.0	59
Notes: This table prese	nts the resu	lts of estimat	ing the relat	tion between	earnings qua	ality and AC	equity com	pensation. Th	e dependent	variables an	e absolute, j	ositive,

Relation between earnings quality and AC equity-based compensation

TABLE 5

All the *t*-statistics are corrected for heteroskedasticity. *, **, and *** denote two-tailed significance levels at 10%, 5%, and 1%, respectively.

⁷⁵⁷

that the mean of AC chair relative power increases from -0.034 in 2007–2009 to 0.013 in 2010–2012 and to 0.106 in 2013–2015. This increasing trend in mean value suggests that AC chairs have become more powerful than CFOs since 2010. Because the CFOs are less likely to have the power to influence firms' earnings management although they may have incentives to do so, the substitution effect between accrual-based and real earnings management may have changed over time.

An alternative explanation for the positive relation between *RESTATE* and *AC_EBC*% is that more effective ACs are more likely to detect and report material misstatements, which would lead to a higher restatement likelihood (Srinivasan et al. 2015). If this alternative explanation is valid, restatements for firms that pay larger equity compensation to their ACs should be less severe if the ACs provide stricter oversight. Following Carcello et al. (2011) and Bruynseels and Cardinaels (2014), we classify a restatement as severe if it involves a change in net income of more than 10% of a firm's beginning total assets in the announcement year. We then regress an indicator for restatement severity on *AC_EBC*% and a set of controls used in prior studies. Untabulated results show that the coefficient on *AC_EBC*% is positive and significant at the 0.10 level (t = 1.72, p < 0.088), suggesting that restatements of firms that pay larger equity compensation to their ACs are more likely to be classified as severe. In fact, the significantly positive coefficients on *AC_EBC*% across our accrual-based and real earnings management measures also provide some evidence that the alternative explanation is not appropriate for interpreting our results.

As in the preceding estimations, the coefficients on BIG4 are insignificant in the discretionary accruals and real earnings management models. These results are consistent with the results in Table 4, which indicate that ACs receiving large equity compensation are less likely to pay a fee premium to their Big 4 auditors. Because the Big 4 auditors do not earn a fee premium, they are less effective at mitigating firms' earnings management. Another possible reason for these insignificant coefficients is that about 96% of our sample firms are audited by Big 4 auditors (Table 2 shows that the mean of BIG4 is 0.964). Due to the small variation in BIG4, the explanatory power of BIG4 in the earnings management models could be weak.¹⁴

Tests of endogeneity

The results in Table 4 show a negative relation between audit fees and AC equity compensation. However, the possibility that this relation is endogenously determined could be a concern in our analysis. The first source of endogeneity is time-invariant unobservable heterogeneity, which arises if there are unobservable firm-specific factors that affect both AC equity compensation and the determination of audit fees. To address this potential endogeneity problem, we use a change model and include firm fixed effects (Kim et al. 2015).

Panel A of Table 6 shows that the coefficient on $\Delta AC_EBC\%$ is negative and significant at the 0.10 level. These results are consistent with the Table 4 results and indicate that the change in equity compensation is negatively related to the change in audit fees. We also re-estimate our regression models after including firm fixed effects. Panel B of Table 6 indicates that the results from the fixed effects regression are consistent with the change analysis results, indicating that the observed negative relation between audit fees and AC equity compensation is robust to controlling for time-invariant unobservable heterogeneity.

The above two approaches may not mitigate the potential endogeneity that could arise when omitted time-variant variables lead to both a change in equity compensation and a change in audit fees. To address this concern, we follow prior research (e.g., Carcello et al. 2002; Vafeas and

^{14.} Recent research using discretionary accruals as measures of audit quality also reports insignificant coefficients on *BIG4*, even though the mean values of *BIG4* are all smaller than ours. For example, the range of the mean value of *BIG4* is between 0.64 and 0.85 in Laurion et al. (2017), Lennox (2016), and Lobo and Zhao (2013). All these studies do not find a significant Big 4 effect in their audit quality models.

TABLE 6

Relation between audit fees and AC	C equity-based	compensation:	Controlling for	or endogeneity
------------------------------------	----------------	---------------	-----------------	----------------

	Panel A:	Change	model	(N =	2,679)
--	----------	--------	-------	------	--------

	$Dep. = Ln(\Delta AF)$		
Variables	Coeff.	<i>t</i> -stat.	
Intercept	0.087	2.79***	
$\Delta AC_EBC\%$	-0.033	-1.83*	
$\Delta SIZE$	0.385	11.84***	
$\Delta FOREIGN$	-0.015	-1.06	
$\Delta SEGMENT$	-0.011	-0.87	
$\Delta CURRENT$	-0.001	-0.21	
$\Delta INVREC$	0.387	2.25**	
ΔROA	-0.114	-1.21	
$\Delta LOSS$	0.010	0.89	
ΔMB	0.002	1.12	
ΔLEV	0.043	0.58	
ΔCFO	-0.815	-2.51**	
$\Delta SPECIAL$	0.005	0.77	
ΔRD	-0.003	-0.05	
$\Delta RESTRUCTURE$	0.011	1.55	
ΔMA	0.009	1.43	
ΔRET_VOL	1.590	1.78*	
ΔICW	0.009	0.24	
ΔCEO_DELTA	0.000	-1.77*	
ΔCEO_VEGA	0.000	-0.68	
ΔCFO_DELTA	0.000	0.33	
ΔCFO_VEGA	0.000	-0.28	
ΔCEO_CHANGE	-0.003	-0.44	
$\Delta BDSIZE$	0.065	1.85*	
$\Delta BDIND$	0.135	1.38	
$\Delta ACSIZE$	-0.020	-1.00	
$\Delta TENURE$	0.063	3.11***	
∆AUDIT_CHANGE	0.040	1.41	
$\Delta BIG4$	0.092	2.05**	
$\Delta LOG(NAS)$	0.004	1.81*	
Industry and Year FE	Ind	cluded	
Adj. R^2	(0.203	

Panel B: Firm fixed effects model (N = 3,685)

	Dep. = Ln(AF)	
Variables	Coeff.	<i>t</i> -stat.
Intercept	11.210	52.07***
AC_EBC%	-0.036	-1.94*
SIZE	0.414	23.40***
FOREIGN	0.006	0.46
SEGMENT	0.009	0.67
CURRENT	-0.014	-2.41**
INVREC	0.474	4.20***

(The table is continued on the next page.)

TABLE 6	(continued)
---------	-------------

Panel B: Firm fixed effects model ($N = 3,685$)			
	Dep. = Ln(AF)		
Variables	Coeff.	<i>t</i> -stat.	
ROA	-0.055	-0.57	
LOSS	0.028	2.19**	
MB	-0.003	-1.93*	
LEV	-0.051	-0.90	
CFO	-0.863	-2.40**	
SPECIAL	0.013	1.56	
RD	0.018	0.45	
RESTRUCTURE	0.019	2.33**	
MA	0.013	1.76*	
RET_VOL	0.854	0.86	
ICW	0.070	2.11**	
CEO_DELTA	0.000	0.00	
CEO_VEGA	0.000	0.03	
CFO_DELTA	0.000	-1.21	
CFO_VEGA	0.000	-0.68	
CEO_CHANGE	0.009	0.89	
BDSIZE	0.128	4.35***	
BDIND	0.000	0.00	
ACSIZE	-0.032	-1.55	
TENURE	0.042	3.11***	
AUDIT_CHANGE	-0.022	-0.74	
BIG4	0.111	2.02**	
LOG(NAS)	0.009	3.76***	
Industry and Year FE	Ind	cluded	
Firm FE	Ind	cluded	
Adj. R ²	C	.969	

Notes: This table presents the results of tests of endogeneity. In panel A, the dependent variable is the change in natural log of total audit fees. In panel B, the models include firm, year and industry fixed effects. See Appendix for variable definitions. All the *t*-statistics are corrected for heteroskedasticity. *, **, and *** denote two-tailed significance levels at 10%, 5%, and 1%, respectively.

Waegelein 2007; Krishnan and Visvanathan 2008) and control for AC ability by including the following three additional variables in model (1): AC multiple directorships (measured as the proportion of AC directors with more than three outside directorships in other companies); tenure of AC members (measured as the proportion of AC directors with board tenure of ten or more years); financial expertise (measured as the proportion of AC members who are financial experts). Because affiliated former partners affect AC effectiveness and enhance audit quality (Christensen et al. 2018), we also include an indicator variable that equals one if at least one AC member is a former partner of the company's audit firm, and zero otherwise. Untabulated results show that former partner affiliation is negatively related to audit fees (t = -1.71, p < 0.089), while the three proxies for AC ability are not. Importantly, the coefficient on $AC_EBC\%$ remains significantly negative at the 0.05 level after including these additional controls, which suggests that our main results are robust to controlling for omitted time-variant variables.

Additional analyses

To provide a sharper view of how specific firm and AC characteristics may accentuate or attenuate AC effectiveness, we examine how the relation between audit fees and AC equity compensation is influenced by the following five characteristics: firm litigation risk, AC bargaining power, AC accounting expertise, audit market competition, and industry expert auditors. We conduct these additional tests by first partitioning the sample into two subsamples based on the annual medians of each of these characteristics and then estimating model (1) separately for each subsample. We summarize the results of these tests in this section and provide more detailed explanation, discussion, and tabulation of results in the online Appendix.

Table A7 in the online Appendix presents the litigation risk results. The table shows that the coefficient on $AC_EBC\%$ is significantly negative at the 0.05 level for the low litigation risk subsample but is insignificant for the high litigation risk subsample. That is, audit fees decrease significantly with increase in AC equity compensation for firms in the low litigation risk subsample but do not change for firms in the high litigation risk subsample. These results suggest that the negative effect of equity compensation on AC audit fee determination does not exist when firms face high litigation risk. Given that ACs are more concerned about litigation risk, relative to ACs of firms that face low litigation risk, ACs of firms that face high litigation risk are likely to be more independent, which would lead to higher audit fees. To the extent that the *high incentive choice* scenario is valid, our results indicate that litigation concerns deter AC directors with high equity compensation from sacrificing independence.

Panel A of Table A8 in the online Appendix presents the AC bargaining power results. The panel shows that for the high AC bargaining power subsample, the coefficient on $AC_EBC\%$ is negative and significant at the 0.05 level. By contrast, the coefficient on $AC_EBC\%$ for the low AC bargaining power subsample is insignificant. Overall, these results suggest that the negative effect of equity compensation on AC audit fee determination does not exist when ACs have low bargaining power. Therefore, the relative bargaining power of the AC significantly changes the observed negative relation between AC equity compensation and audit fees.

Panel B of Table A8 in the online Appendix presents the AC accounting expertise results. The panel shows that unlike the coefficient on $AC_EBC\%$ for the low accounting expert subsample, which is negative and significant at the 0.05 level, the corresponding coefficient for the high accounting expert subsample is positive and insignificant. These results suggest that accounting expertise can mitigate the impairment of AC independence due to large equity compensation, and this mitigating effect occurs only when accounting experts represent a high proportion of AC membership. Although prior research provides extensive evidence that AC accounting expertise enhances financial reporting quality (e.g., Bedard et al. 2004; Dhaliwal et al. 2010; Krishnan and Visvanathan 2008), there is a lack of research on whether accounting expertise can mitigate the adverse effect of equity compensation by improving AC independence during the fee determination process. We contribute to the literature by providing empirical evidence on this issue.

Panel C of Table A8 in the online Appendix presents the audit market competition results. The panel shows that the coefficient on $AC_EBC\%$ is insignificant in the low market competition subsample but is negative and significant at the 0.05 level in the high market competition subsample. These results indicate that the observed negative relation between AC equity compensation and audit fees occurs only when audit market competition is high. In terms of economic significance, for a one standard deviation increase in $AC_EBC\%$, the audit fees will decrease by \$68,361 (i.e., $-0.096 \times 0.207 \times \exp(15.051)$) in the high competition subsample. Importantly, this reduction in audit fees is larger than the fee reduction reported in Table 4 (i.e., \$59,104). These results suggest that the negative effect of AC equity compensation on audit fees is stronger when audit market competition is higher. Whereas prior studies report that audit market competition is positively associated with restatements (e.g., Numan and Willekens 2012), our results suggest that ACs receiving larger equity compensation appear to take advantage of the high market

competition to reduce audit fees to an even lower level. We thus contribute to the literature by providing evidence on how audit market competition and AC equity compensation jointly influence AC audit fee decisions.

Table A9 in the online Appendix presents the industry expert auditor results. The table shows that the coefficient on $AC_EBC\%$ is negative and significant at the 0.05 level in the nonindustry expert subsample, which is consistent with our main results in Table 4. This coefficient is insignificant in the industry expert subsample, which suggests that the negative effect of equity compensation on audit fees is mitigated when auditors are industry experts. Two reasons may explain this finding. First, industry expertise may increase AC concerns that reducing audit fees could draw auditors' attention to AC independence being compromised. Second, industry expertise may increase auditors' negotiation power in determining audit fees (Casterella et al. 2004). In fact, the results reported in Table A9 are consistent with those reported in panel A of Table A8, which show that the coefficient on $AC_EBC\%$ is insignificant when auditors have high bargaining power. Unlike prior studies that examine the relation between auditor industry expertise and audit fees from a supply-side perspective and provide mixed results, we contribute to the literature by investigating how demand-side ACs react to industry expertise and showing that industry expertise appears to protect auditors from a reduction in audit fees by compromised ACs.

Robustness tests

We briefly summarize the results of robustness tests in this section and provide more detailed explanation, discussion, and tabulation of results in the online Appendix.

First, we reestimate our audit fee model (1) after controlling for differences in firm size and find that the coefficient on AC_EBC% is negative and significant at the 0.05 level (panel B of Table A10 in the online Appendix). Second, because firms that do not offer equity compensation to their ACs may have different compensation policies and governance structures, we reestimate model (1) after excluding 154 firm-year observations (see panel A of Table 1) with no AC equity compensation and find that the coefficient on $AC_EBC\%$ is negative and significant at the 0.10 level (panel A of Table A11 in the online Appendix). Third, we use several alternative measures of AC equity compensation in model (1) and find negative and significant coefficients on these alternative measures (panel B of Table A11 in the online Appendix). Fourth, as a placebo test, we reestimate model (1) after replacing AC equity compensation with AC cash compensation and find an insignificant coefficient on AC cash compensation (panel B of Table A11 in the online Appendix). Fifth, we do not truncate extreme values and use robust regression based on the MMestimation to estimate model (1). We find that the coefficient on AC_EBC% is negative and significant at the 0.05 level (Table A12 in the online Appendix). Sixth, to examine whether our results are sensitive to including the financial crisis period, we reestimate model (1) using firmyear observations during the 2011-2015 postcrisis period. Again, untabulated results show that the coefficient on AC_EBC% is negative and significant at the 0.05 level.

5. Summary and conclusions

The role of ACs in enhancing financial reporting quality has received much attention over the past decade. Therefore, evidence on factors that may influence the effectiveness of ACs is of interest to regulators, boards of directors, and investors. Although regulators and the press have expressed concerns about whether equity compensation weakens AC oversight effectiveness, extant literature provides only limited and mixed results on this issue. In this study, we examine whether AC equity compensation enhances or impairs AC determination of audit fees. Using a sample of 3,685 firm-year observations from 2007 to 2015, we find evidence suggesting that equity compensation induces AC members to compromise their independence by paying lower audit fees. We further show that larger equity compensation is associated with lower earnings quality. These results are robust to controlling for endogeneity, firm size differences, alternative

measures of equity compensation, alternative treatment of extreme values, and the effect of the financial crisis.

We further examine how the relation between audit fees and AC equity compensation is influenced by certain firm, AC, and auditor characteristics. We find that the negative effect of equity compensation on AC independence no longer manifests when (i) firms face high litigation risk, (ii) auditors have stronger bargaining power in negotiating audit fees, (iii) most of the AC members are accounting experts, and (iv) auditors are industry experts.

Our results have important implications for regulators, boards of directors, investors, and others. First, although the appointment of auditors and the determination of audit fees are generally viewed as solutions to enhance financial reporting quality, our results indicate that choosing an appropriate level of equity compensation for ACs is also important. Second, whereas SOX and the SEC only require directors to be fully independent before they become AC members, our findings highlight the need for regulatory restrictions on compensation practices so that AC members can maintain independence in fact rather than in appearance and fulfill their duties in determining audit fees. Third, even though we find that equity compensation attenuates AC independence in determining appropriate audit fees, we also find that this negative effect does not hold when the auditors have higher bargaining power and the ACs include more accounting experts. Regulators may consider new regulations that more evenly balance the bargaining power between the ACs and the auditors (e.g., mandatory audit firm rotation) and strengthen the role of accounting experts in ACs. Fourth, regulators have debated how audit market competition influences audit quality. Whereas some argue that high competition is desirable because market consolidation may cause audit quality to decrease (e.g., US Chamber of Commerce 2006), others have expressed concern that high audit market competition motivates opinion shopping and thereby decreases audit quality (PCAOB 2011). We inform this debate by providing evidence that audit market competition and AC equity compensation may jointly affect AC audit fee decisions and the resulting earnings quality. Fifth, while we focus on equity compensation as one major factor that could harm AC effectiveness, future research could examine factors (e.g., tenure of membership, penalties for oversight failures) that may also affect other aspects of AC effectiveness.

Appendix

Variable definitions

Variables	Definitions
Dependent var	iables
Ln(AF)	Natural log of total audit fees
Abs_DA	Absolute value of a firm's performance-adjusted discretionary accruals. We follow Kothari et al. (2005) and estimate performance-adjusted discretionary accruals by including return on assets (<i>ROA</i>) in the following model:
	$TA_{it} = \delta_0 + \delta_1 \left(\frac{1}{ASSETS_{it-1}}\right) + \delta_2 \Delta SALES_{it} + \delta_3 PPE_{it} + \delta_4 ROA_{it} + \varepsilon_{it},$
	where TA_{it} is total assets, $\Delta SALES_{it}$ is change in sales divided by lagged total assets ($ASSETS_{it-1}$), PPE_{it} is net property, plant, and equipment divided by $ASSETS_{it-1}$, and ROA_{it} is return on assets
RESTATE	An indicator that equals one for firm-years in which a firm's reported earnings are restated, and zero otherwise
REM1	An aggregate measure of real activities management calculated as the sum of <i>RDISX</i> and <i>RPROD</i> . <i>RDISX</i> is the level of abnormal discretionary expenses measured as the residuals from the following industry-year regression:

(The Appendix is continued on the next page.)

Appendix (continued)

Variables	Definitions
	$\frac{DISX_{it}}{Assets_{i,t-1}} = k_0 + k_1 \frac{1}{Assets_{i,t-1}} + k_2 \frac{Sales_{it-1}}{Assets_{i,t-1}} + \varepsilon_{it},$
	where <i>DISX</i> is discretionary expenses (the sum of advertising expenses, R&D expenses and SG&A expenses) during the year. We multiply the residuals by -1 . <i>RPROD</i> is the level of abnormal production costs measured as the residuals from the following industry-year regression: $\frac{PROD_{il}}{Assets_{i,l-1}} = k_0 + k_1 \frac{1}{Assets_{i,l-1}} + k_2 \frac{Sales_{il}}{Assets_{i,l-1}} + k_3 \frac{\Delta Sales_{i,l-1}}{Assets_{i,l-1}} + \epsilon_{it},$
	where PROD is production costs (the sum of costs of goods sold and change in
REM2	inventory) during the year An aggregate measure of real activities management calculated as the sum of <i>RCFO</i> and <i>RDISX. RCFO is</i> the level of abnormal cash flow from operations measured as the residuals from the following industry-year regression: $\frac{CFO_{it}}{Assets_{it-1}} = k_0 + k_1 \frac{1}{Assets_{it-1}} + k_2 \frac{Sales_{it}}{Assets_{it-1}} + k_3 \frac{\Delta Sales_{it}}{Assets_{it-1}} + \varepsilon_{it},$
	where <i>CFO</i> is cash flow from operations, <i>Sales</i> is annual sales, and <i>Assets</i> is total
Test veriable	assets. We multiply the residuals by -1
AC_EBC%	Average of the ratio of equity compensation to total compensation across all members of a firm's ΔC
Control variables	of a min s AC
Supply-side fee dete	rminants
(1) Firm's operation	complexity
SIZE	Natural log of total assets
FOREIGN	An indicator that equals one if a firm's foreign assets are greater than 0, and zero otherwise
SEGMENT	Natural log of the number of a firm's business and geographic segments
(2) Firm's operating	risk
CURRENT	Current assets / Current liabilities
ROA	Net income before taxes and extraordinary items / Average total assets
LOSS	An indicator that equals one if a firm's net income is less than 0, and zero otherwise
MB	Market to book ratio
(3) Firm's inherent	risk
INVREC	Sum of accounts receivable and inventory / Total assets.
(4) Firm's financial	risk
LEV	Total debt / Total assets
CFO	Cash flow from operations / Average assets
SPECIAL	An indicator variable that equals one if the firm reports special items in the income statement, and zero otherwise
RD	An indicator variable that equals one if the firm reports research and development expenses, and zero otherwise
RESTRUCTURE	An indicator variable that equals one if the firm was involved in a debt restructuring, and zero otherwise
MA	An indicator variable that equals one if the firm was involved in a merger or acquisition activity, and zero otherwise

(The Appendix is continued on the next page.)

Appendix (continued)

Variables	Definitions
RET_VOL	Standard deviation of firm's market-adjusted monthly stock returns measured over the previous 60 months
(5) Firm's control ris	sk
ICW	An indicator variable that equals one if a firm's auditor indicates material internal control weaknesses, and zero otherwise
(6) Audit risk	
CEO_DELTA	Change in the dollar value of the CEO's equity holdings for a 1% change in stock price
CEO_VEGA	Change in the dollar value of the CEO's equity holdings for a 1% change in the firm's stock return volatility
CFO_DELTA	Change in the dollar value of the CFO's equity holdings for a 1% change in stock price
CFO_VEGA	Change in the dollar value of the CFO's equity holdings for a 1% change in the firm's stock return volatility
CEO_CHANGE	An indicator variable that equals one if a firm changes its CEO, and zero otherwise
Demand-side fee det	erminants
BDSIZE	Natural log of the number of board members
BDIND	Percentage of independent directors on the board
ACSIZE	Natural log of the number of AC members
TENURE	Natural log of the auditor's tenure
AUDIT_CHANGE	An indicator variable that equals one if the firm changes its auditor, and zero otherwise
BIG4	An indicator that equals one if a firm is audited by a Big 4 auditor, and zero otherwise
LOG(NAS)	Natural log of total nonaudit service fees plus one

References

- Adams, R. B., and D. Ferreria. 2009. Do directors perform for pay? *Journal of Accounting and Economics* 46 (1): 154–71.
- Alexander, J. 1991. Do the merits matter? A study of settlements in securities class actions. Stanford Law Review 43 (3): 497–598.
- Archambeault, F. T. D., and D. Hermanson. 2008. Audit committee incentive compensation and accounting restatements. *Contemporary Accounting Research* 25 (4): 965–92.
- Armstrong, C., D. Larcker, G. Ormazabal, and D. Taylor. 2013. The relation between equity incentives and misreporting: The role of risk-taking incentives. *Journal of Financial Economics* 109 (2): 327–50.
- Audit Committee Collaboration. 2013. Enhancing the audit committee report: A call to action, retrieved November 20, 2013, https://www.thecaq.org/enhancing-audit-committee-report-call-action/.
- Austin, P., A. Rojik, and C. Tower. 2018. Voluntary audit committee disclosures continue to evolve. Chicago, IL: BDO LLP.
- Badolato, P., D. Donelson, and M. Ege. 2014. Audit committee financial expertise and earnings management: The role of status. *Journal of Accounting and Economics* 58 (2–3): 208–30.
- Beardsley, L., D. Lassila, and T. Omer. 2019. How do audit offices respond to audit fee pressure? Evidence of increased focus on nonaudit services and their impact on audit quality. *Contemporary Accounting Research* 36 (2): 999–1027.
- Beasley, M. S. 1996. An empirical analysis of the relation between the board of director composition and financial fraud. *The Accounting Review* 71 (2): 443–65.

- Beasley, M., J. Carcello, D. Hermanson, and T. Neal. 2009. The AC oversight process. *Contemporary* Accounting Research 26 (1): 65–122.
- Beck, M. J., and E. G. Mauldin. 2014. Who's really in charge? Audit committee versus CFO power and audit fees. *The Accounting Review* 89 (6): 2057–85.
- Bedard, J. C., S. M. Chtourou, and L. Courteau. 2004. The effect of audit committee expertise, independence, and activity on aggressive earnings management. *Auditing: A Journal of Practice & Theory* 23 (2): 13–35.
- Billings, B., X. Gao, and Y. Jia. 2014. CEO and CFO equity incentives and the pricing of audit services. Auditing: A Journal of Practice & Theory 33 (2): 1–25.
- Bills, K. L., D. C. Jeter, and S. E. Stein. 2015. Auditor industry specialization and evidence of cost efficiencies in homogenous industries. *The Accounting Review* 90 (5): 1721–54.
- Bruynseels, L., and E. Cardinaels. 2014. The audit committee: Management watchdog or personal friend of the CEO? *The Accounting Review* 89 (1): 113–45.
- Cairney, T. D., and E. G. Stewart. 2015. Audit fees and client industry homogeneity. Auditing: A Journal of Practice and Theory 34 (4): 33–57.
- Campbell, J. L., J. Hansen, C. A. Simon, and J. L. Smith. 2015. Audit committee stock options and financial reporting quality after the Sarbanes-Oxley Act of 2002. *Auditing: A Journal of Practice & Theory* 34 (2): 91–120.
- Carcello, J., D. Hermanson, T. Neal, and R. Riley. 2002. Board characteristics and audit fees. *Contemporary Accounting Research* 19 (1): 365–84.
- Carcello, J. V., D. R. Hermanson, and Z. Ye. 2011. Corporate governance research in accounting and auditing: Insights, practice implications, and future research directions. *Auditing: A Journal of Practice & Theory* 30 (3): 1–31.
- Casterella, J. R., J. R. Francis, B. L. Lewis, and P. L. Walker. 2004. Auditor industry specialization, client bargaining power, and audit pricing. *Auditing: A Journal of Practice & Theory* 23 (1): 123–40.
- Center for Audit Quality (CAQ). 2015. External auditor assessment tool: A reference for US audit committees, retrieved June 2, 2015, https://www.thecaq.org/external-auditor-assessment-tool-reference-usaudit-committees.
- Chen, Y., F. A. Gul, M. Veeraraghavan, and L. Zolotoy. 2015. Executive equity risk-taking incentives and audit pricing. *The Accounting Review* 90 (6): 2205–34.
- Chi, W., L. L. Lisic, and M. Pevzner. 2011. Is enhanced audit quality associated with greater real earnings management? Accounting Horizons 25 (2): 315–35.
- Choi, J.-H., J.-B. Kim, and Y. Zang. 2010. Do abnormally high audit fees impair audit quality? Auditing: A Journal of Practice & Theory 29 (2): 115–40.
- Christensen, B. E., T. C. Omer, M. K. Shelley, and P. A. Wong. 2018. Affiliated former partners on the audit committee: Influence on the auditor–client relationship and audit quality. *Auditing: A Journal of Practice & Theory* 38 (3): 95–119.
- Cohen, D., A. Day, and T. Lys. 2008. Real and accrual-based earnings management in the pre- and post-Sarbanes-Oxley periods. *The Accounting Review* 83 (3): 757–87.
- Core, J. E., and R. W. Guay. 2002. Estimating the value of employee stock option portfolios and their sensitivities to price and volatility. *Journal of Accounting Research* 40 (3): 613–30.
- Corporate Finance Institute. 2020. *Big Four accounting salary—An overview of accounting compensation at the Big Four*, retrieved 1 June, 2020, https://corporatefinanceinstitute.com/resources/careers/compensation/big-four-accounting-salary/
- Dalton, D. R., C. M. Daily, T. Certo, and R. Roengpitya. 2003. Meta-analyses of financial performance and equity: Fusion or confusion? Academy of Management Journal 46 (1): 13–26.
- DeFond, M. L., and J. R. Francis. 2005. Does auditing matter? *Auditing: A Journal of Practice & Theory* 24 (Supplement): 5–30.
- DeFond, M., and J. Y. Zhang. 2014. A review of archival auditing research. Journal of Accounting and Economics 58 (2–3): 275–326.

- Dhaliwal, D., V. Naiker, and F. Navissi. 2010. The association between accrual quality and the characteristics of accounting experts and mix of expertise on audit committee. *Contemporary Accounting Research* 27 (3): 787–827.
- Doty, J. 2014. PCAOB concerned about fee decreases. *Accounting Today*, retrieved May 1, 2014, http:// www.accountingtoday.com/news/auditing/pcaob-concerned-about-audit-fee-decreases-70530-1.html.
- Engel, E., R. Hayes, and X. Wang. 2010. Audit committee compensation and the demand for monitoring of the financial reporting process. *Journal of Accounting and Economics* 49 (1/2): 136–54.
- Ernst & Young. 2017. Audit committee reporting to shareholders in 2017. Center for Board Matters, http:// www.ey.com/Publication/vwLUAssets/ey-audit-committee-reporting-to-shareholders-in-2017/\$FILE/ ey-audit-committee-reporting-to-shareholders-in-2017.pdf
- Fama, E. F. 1980. Agency problems and the theory of the firm. *Journal of Political Economy* 88 (2): 288–307.
- Fama, E. F., and M. C. Jensen. 1983. Separation of ownership and control. *Journal of Law and Economics* 26 (2): 301–25.
- Ferguson, A., J. R. Francis, and D. J. Stokes. 2003. The effects of firm-wide and office-level industry expertise on audit pricing. *The Accounting Review* 78 (2): 429–48.
- Francis, J., K. Reichelt, and D. Wang. 2005. The pricing of national and city-specific reputations for industry expertise in the US audit market. *The Accounting Review* 80 (1): 113–36.
- Gong, Q. H., O. Z. Li, Y. P. Lin, and L. S. Wu. 2016. On the benefits of audit market consolidation: Evidence from merged audit firms. *The Accounting Review* 91 (2): 463–88.
- Hillman, A., and T. Dalziel. 2003. Boards of directors and firm performance: Integrating agency and resource dependence perspectives. *Academy of Management Review* 28 (3): 383–96.
- Hogan, C. E., and M. S. Wilkins. 2008. Evidence on the audit risk model: Do auditors increase audit fees in the presence of internal control deficiencies? *Contemporary Accounting Research* 25 (1): 219–42.
- Jayaraman, S., and T. Milbourn. 2009. Does equity-based CEO compensation really increase litigation risk? Working Paper, Washington University at St. Louis.
- Jensen, M. C., and W. H. Meckling. 1976. Theory of the firm, managerial behavior, agency costs and ownership structure. *Journal of Financial Economics* 3 (1): 305–60.
- Jensen, M. C., and R. S. Ruback. 1983. The market for corporate control: The scientific evidence. Journal of Financial Economics 11 (1/4): 5–50.
- Keune, M. B., and K. M. Johnstone. 2015. Audit committee incentives and the resolution of detected misstatements. Auditing: A Journal of Practice & Theory 34 (4): 109–37.
- Kim, I., and D. J. Skinner. 2012. Measuring securities litigation risk. *Journal of Accounting and Economics* 53 (1–2): 290–310.
- Kim, Y., H. Li, and S. Li. 2015. CEO equity incentives and audit fees. *Contemporary Accounting Research* 32 (2): 608–38.
- Kinney, W. R., Jr., and R. Libby. 2002. Discussion of the relation between auditors' fees for nonaudit services and earnings management. *The Accounting Review* 77 (S-1): 107–14.
- Kothari, S. P., A. J. Leone, and C. E. Wasley. 2005. Performance matched discretionary accrual measures. *Journal of Accounting and Economics* 39 (1): 163–97.
- KPMG. 2006. Audit committees feel most exposed to litigation, http://www.accountingeducation.com/index. cfm?page=newsdetails&id=143072
- Krishnan, G., and G. Visvanathan. 2008. Does the SOX definition of an accounting expert matter? The association between audit committee directors' accounting expertise and accounting conservatism. *Contemporary Accounting Research* 25 (3): 827–57.
- Lambert, R. A., C. Leuz, and R. E. Verrecchia. 2007. Accounting information, disclosure, and the cost of capital. *Journal of Accounting Research* 45 (2): 385–420.
- Laurion, H., A. Lawrence, and J. P. Ryans. 2017. US audit partner rotations. *The Accounting Review* 92 (3): 209–37.
- Lennox, C. S. 2016. Did the PCAOB's restrictions on auditors' tax services improve audit quality? *The Accounting Review* 91 (5): 1493–512.

- Leone, A. J., M. Minutti-Meza, and C. E. Wasley. 2019. Influential observations and inference in accounting research. *The Accounting Review* 94 (6): 337–64.
- Lobo, G. J., and Y. Zhao. 2013. Relation between audit effort and financial report misstatements: Evidence from quarterly and annual restatements. *The Accounting Review* 88 (4): 1385–412.
- Magilke, M., W. Mayhew, and J. Pike. 2009. Are independent audit-committee members objective? Experimental evidence. *The Accounting Review* 84 (6): 1959–81.
- Menon, K., and D. D. Williams. 1994. The insurance hypothesis and market prices. *The Accounting Review* 69 (2): 327–42.
- Millstein, I. M. 2002. Oversight Hearing on Accounting and Investor Protection Issues Raised by Enron and Other Public Companies. Washington, DC: US Senate Committee on Banking, Housing, and Urban Affairs.
- Monks, R. A., and N. Minow. 2011. Corporate governance. 5th ed.New York: John Wiley & Sons.
- Naiker, V., and D. Sharma. 2009. Former audit partners on the audit committee and internal control deficiencies. *The Accounting Review* 84 (2): 559–87.
- Naiker, V., D. S. Sharma, and V. D. Sharma. 2013. Do former audit firm partners on audit committees procure greater nonaudit services from the auditor? *The Accounting Review* 88 (1): 297–326.
- National Association of Corporate Directors (NACD). 2015. NACD 2014–2015 Director Compensation Report. Washington, DC: NACD.
- Neter, J. 1996. Applied linear statistical models. Boston, MA: McGraw-Hill.
- Newton, N. J., J. S. Persellin, D. C. Wang, and M. S. Wilkins. 2016. Internal control opinion shopping and audit market competition. *The Accounting Review* 91 (2): 603–23.
- Numan, W., and M. Willekens. 2012. An empirical test of spatial competition in the audit market. *Journal* of Accounting and Economics 53 (1–2): 450–65.
- Petersen, M. 2009. Estimating standard errors in finance panel data sets: Comparing approaches. *Review of Financial Studies* 22 (1): 435–80.
- PCAOB. 2010a. Report on observations of PCAOB inspectors related to audit risk areas affected by the economic crisis. Washington, DC: PCAOB.
- PCAOB. 2010b. Auditing standards related to the auditor's assessment of and response to risk and related amendments to PCAOB Standards. PCAOB Release No. 2010–004. Washington, DC: PCAOB.
- PCAOB. 2010c. Auditing Standard No. 2110: Identifying and assessing risks of material misstatement. PCAOB Release No. 2010-004. Washington, DC: PCAOB.
- PCAOB. 2011. Concept release on auditor independence and audit firm rotation. PCAOB Release No. 2011-006. Washington, DC: PCAOB.
- Reichelt, K. J., and D. Wang. 2010. National and office-specific measures of auditor industry expertise and effects on audit quality. *Journal of Accounting Research* 48 (3): 647–86.
- SEC. Standards related to listed company audit committees, retrieved April 25, 2003, from http://www.sec. gov/rules/final/33-8220.htm.
- SEC. *Possible revisions to audit committee disclosures*, retrieved September 8, 2016, from https://www.sec. gov/rules/concept/2015/33-9862.pdf.
- Seetharaman, A., F. Gul, and S. Lynn. 2002. Litigation risk and audit fees: Evidence from UK firms crosslisted on US markets. *Journal of Accounting and Economics* 33 (1): 1–115.
- Sengupta, P., and S. Zhang. 2015. Equity-based compensation of outside directors and corporate disclosure quality. *Contemporary Accounting Research* 32 (3): 1073–98.
- Shleifer, A., and R. W. Vishny. 1989. Management entrenchment: The case of managers' specific investments. *Journal of Financial Economics* 25 (1): 123–39.
- Srinivasan, S. 2005. Consequences of financial reporting failure for outside directors: Evidence from accounting restatements and AC members. *Journal of Accounting Research* 43 (2): 291–334.
- Srinivasan, S., A. S. Wahid, and G. Yu. 2015. Admitting mistakes: Home country effect on the reliability of restatement reporting. *The Accounting Review* 90 (3): 1201–40.
- US Chamber of Commerce. 2006. Auditing: A profession at risk. Washington, DC: US Chamber of Commerce.

- Vafeas, N., and J. Waegelein. 2007. The association between audit committees, compensation incentives, and corporate audit fees. *Review of Quantitative Finance and Accounting* 28 (3): 241–55.
- White, M. J. 2014. Remarks at the financial accounting foundation trustees dinner. SEC Speech. May 20, https:// www.sec.gov/news/speech/2014-spch052014mjw
- Whitehouse, T. 2014. SEC plans early 2015 concept release on audit committee upgrades. *Compliance Week*, retrieved October 22, 2014.
- Zang, A. Y. 2012. Evidence on the trade-off between real activities manipulation and accrual-based earnings management. *The Accounting Review* 87 (2): 675–703.

SUPPORTING INFORMATION

Additional supporting information may be found in the online version of this article: **Online Appendix:** Supporting information